Author's response to reviews

Title: A systematic review of psychosocial interventions for family carers of palliative care patients

Authors:

Peter L Hudson (peterh@medstv.unimelb.edu.au)
Cheryl Remedios (cheryl.remedios@svhm.org.au)
Kristina Thomas (Kristina.Thomas@svhm.org.au)

Version: 3 Date: 21 July 2010

Author's response to reviews:

21 July 2010

Miss Nafisa Qazi
The BioMed Central Editorial Team

Dear Miss Qazi

Re: MS: 1086809723319261
A systematic review of psychosocial interventions for family carers of palliative care patients. Authors: Peter L Hudson, Cheryl Remedios, Kristina Thomas

Thank you for your email of 20 July 2010.

The revised manuscript is attached including:
• track change of amended typographical error,
• author affiliations,
• competing interest section,
• authors' contributions section,
• acknowledgements.

In response to Reviewer, Ann-Louise Caress' comment that there is a typographical error on the first line of the last paragraph (caregivers), I have amended the manuscript accordingly.

Please note author affiliations as follows:
• Peter L Hudson, Centre for Palliative Care c/o St Vincent's Hospital and The University of Melbourne, Australia and Queens University, Belfast
• Chery Remedios, Centre for Palliative Care c/o St Vincent's Hospital and The University of Melbourne, Australia
• Kristina Thomas, Centre for Palliative Care c/o St Vincent's Hospital and The
University of Melbourne, Australia

I trust that you will find all of these improvements to the manuscript satisfactory.

Yours sincerely

Prof Peter Hudson